Author's response to reviews

Title: Cardiac arrest provoked by itraconazole and amiodarone interaction: a case report

Authors:

Angeliki M Tsimogianni Dr (atsimog@med.uoa.gr)
Ilias Andrianakis Dr (iandrianakis@yahoo.gr)
Alex Betrosian Dr (abetrosian@gmail.com)
Emmanouil Douzinas Dr (edouzin@cc.uoa.gr)

Version: 6 Date: 2 April 2011

Author's response to reviews:

2nd April 2011

Dear Editor in Chief of Journal of Medical Case Reports

We highly appreciate the constructive critiques of our manuscript MS: 4902778904510298 entitled “Cardiac arrest provoked by itraconazole and amiodarone interaction: a case report”, as well as the opportunity to submit a revised version of the paper. We have undertaken every effort feasible to address the grammatical mistakes raised by the associate editor in a revised version of the manuscript (MS: 4902778904510298.R3). As a result, we believe that the manuscript is substantially improved and hope that it is acceptable for publication. Below is a point-by-point response (R; in red) to the editor comments (C; in italics).

Sincerely,

Angeliki Tsimogianni, MD, PhD, Pneumologist
3rd Intensive Care Department, Evgenidion Hospital, 20 Papadiamantopoulou street, 11528, Athens, Greece. Work Tel: +30 210 7208116; Home Tel: +30 210 9944013; Fax: +30-210-7220540.
E-mail: atsimog@med.uoa.gr.

ASSOCIATE EDITOR COMMENTS TO AUTHOR

This manuscript will still need revision. It reads better than last time but there are still grammatical mistakes within the manuscript. Whilst it has good educational value, these will need addressing before it can be published. Once these minor mistakes are corrected, I am happy to accept the manuscript.

Comments

C. Page 5.
'In the coming hours, patient’s level of consciousness deteriorated and required intubation and transfer to ICU.'

To change to 'In the coming hours, patient’s level of consciousness deteriorated and transferred to ICU.'

R1. Page 5, The sentence has been corrected as suggested by the editor.


'The patient was connected to ventilator again, cardiac massage for 3 minutes was performed and 2mg of adrenaline were administered to regain pulses.'

To change to 'The patient was connected to ventilator again, cardiac massage was performed for 3 minutes and 2mg of adrenaline were administered until the return of spontaneous circulation'.

R2. Page 6. We thank the editor for his suggestion and we made the relevant changes to the manuscript.

C3. ‘arrested again’ to change to suffered another PEA arrest (assuming it was pulseless electrical activity)?

R3. Page 6. We agree with the editor that suffered another PEA arrest is a nicer and more accurate expression and we made the correction in the manuscript as well as in the abstract (page 2).

Declaration of competing interests:
I declare that I have no competing interests